Author’s response to reviews

Title: The Magnitude and correlates of Esophageal Varices among newly diagnosed Cirrhotic Patients Undergoing Screening Upper Gastrointestinal Endoscopy before incident bleeding in North-Western Tanzania; a Cross Sectional Study

Authors:
Daniel Gunda (daniel_rev2002@yahoo.com)
Semvua Kilonzo (sekipcb@yahoo.com)
Zakhia Mamballah (zakyzarin@gmail.com)
Paulina Manyiri (paulinamanyi@yahoo.com)
David Majinge (dav_jingeson@yahoo.com)
Hyasinta Jaka (yasintaliwa5@gmail.com)
Benson Kidenya (benkidenya@yahoo.com)
Humphrey Mazigo (humphreymazigo@gmail.com)

Version: 4 Date: 02 Nov 2019

Author’s response to reviews:

The editor in chief
BMC Gastroenterology

Dear Sir/Madam;

RE: SUBMISSION OF A FINAL REVISED VERSION OF MANUSCRIPT WITH ID: BMGE-D-18-00776R3

We pleased to submit a final version of our manuscript mentioned above for possible consideration of publication in your journal. Below is a point response to the editor’s comments

1. Your manuscript needs to be copyedited. We recommend you use a professional language editing service. RESPONSE: A detailed language editing has been done and has made this final version a much better one
2. Please provide a list of all the abbreviations used in the manuscript. This list should be placed just before the Declarations section. All abbreviations should still be defined in the text at first use.

RESPONSE: the list of abbreviation is given just under the declarations part, and they are also defined in the text at their first use

3. We note that you have not included a ‘Funding’ section in the Declarations.

RESPONSE: Funding statement is added under declaration accordingly

4. In order to be in line with journal requirements, please include the following headings within your manuscript: Background, Methods, Results, Discussion, Conclusions

RESPONSE: The manuscript is formatted into abstract, Background, Methods, Results, Conclusions, Declarations, abbreviations, references and tables/figures

5. In your "ethical approval and consent to participate" section of your declarations, please clarify whether the consent obtained was written or verbal.

RESPONSE: The clarification is done accordingly. The consent obtained was an informed written one

Best regards

_______________________
Daniel W. Gunda, MD, Mmed, PhD candidate
daniel_rev2002@yahoo.com
+255757766222